Author’s response to reviews

Title: Severe Skin Disease in Lupus associated with Hemophagocytic Lymphohistiocytosis: Case Reports and Review of the Literature

Authors:

Christina Thornton (christina.thornton@albertahealthservices.ca)
Parham Minoo (Parham.Minoo@cls.ab.ca)
Michelle Schneider (Michelle.Schneider@cls.ab.ca)
Aurore Fifi-Mah (Aurore.Fifi-Mah@albertahealthservices.ca)

Version: 2 Date: 20 Dec 2018

Author’s response to reviews:

Thank you for the comments. Please see below the point by point changes:

1. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and before the References.
   - Abbreviations
   - Ethics approval and consent to participate
   - Consent for publication
   - Availability of data and material
   - Competing interests
   - Funding
   - Authors' contributions
   - Acknowledgements
   - Thank you. This has been revised between lines 234-252. Of note funding was not applicable and as such was indicated

2. Please rename "Discussion" to "Discussion and Conclusions".
- Thank you. This has been changed on line 156.

3. Please relocate the Figures List after the references.

- Thank you. This has been relocated after the references on line 327.

4. Thank you for providing your CARE checklist. However, at this stage it is not required and so we kindly ask that you remove it from your manuscript.

- Thank you. This has been removed.